Author’s response to reviews

Title: Biased Technical Change in Hospital Care and the Demand for Physicians

Authors:

Jos Blank (j.blank@ipsestudies.nl)

Thomas K. Niaounakis (t.niaounakis@ipsestudies.nl)

Vivian G. Valdmanis (vivianvaldmanis@gmail.com)

Version: 1 Date: 12 Jun 2020

Author’s response to reviews:

We have replaced the manuscript by a revised version. This includes a Discussion section and a list of abbreviations. We changed the order of the List of abbreviations.